Author’s response to reviews

Title: Developing and evaluating interventions to reduce inappropriate prescribing by general practitioners of antibiotics for upper respiratory tract infections: a randomised controlled trial to compare paper-based and web-based modelling experiments

Authors:

Shaun Treweek (streweek@mac.com)
Ian W Ricketts (ricketts@computing.dundee.ac.uk)
Jillian Francis (j.francis@abdn.ac.uk)
Martin Eccles (Martin.eccles@ncl.ac.uk)
Debbie Bonetti (d.bonetti@cpse.dundee.ac.uk)
Nigel B Pitts (n.b.pitts@cpse.dundee.ac.uk)
Graeme MacLennan (g.maclennan@abdn.ac.uk)
Frank Sullivan (f.m.sullivan@cpse.dundee.ac.uk)
Claire Jones (cjones@computing.dundee.ac.uk)
Mark Weal (mjw@ecs.soton.ac.uk)
Karen Barnett (k.barnett@cpse.dundee.ac.uk)

Version: 5 Date: 24 February 2011

Author’s response to reviews: see over
Dear Editor,

Re: Implementation Science manuscript: 1720910543500656

I have uploaded a modified version of our protocol. Specifically I have done the following

- matched the author names in the manuscript to those in the submission system
- checked the email address and corrected that of one author to match the submission system
- added the trial registration number to the abstract held in the submission system.
- added the full list of authors to each of the articles listed in the References.
- provided full citations for References 8, 9 and 19.
- removed the additional files form the submission system. After a telephone conversation with one of your staff, we agreed that the additional documents submitted with the protocol (letter from the ethics committee, award letter and CONSORT checklist) were needed for the editorial process not the published article.
- removed as much white space from Figure 1 as I could.

I look forward to hearing from you in due course.

Yours faithfully,

Dr Shaun Treweek
Senior Lecturer, University of Dundee
Assistant Director, Tayside Clinical Trials Unit

Contact details:
Division of Clinical & Population Science and Education
University of Dundee
Kirsty Semple Way
Dundee DD2 4BF
UK